

ORIGINAL PAPER

# Joint custody law and mothers' labor market outcomes: evidence from the USA

Daniela Vuri<sup>1</sup> 🝺

Received: 20 May 2016 / Accepted: 13 November 2017 / Published online: 8 December 2017 © Springer-Verlag GmbH Germany, part of Springer Nature 2017

**Abstract** This paper studies the economic implications for mothers of the changes in child custody law from maternal preference to joint custody using the 1960–2000 Census Public Use Micro Sample (IPUMS). Variation in the timing of the joint custody reform across states provides a natural experimental framework to study the causal effect of shared custody on mothers' economic outcomes. The results show that only single mothers experience a decrease in earnings as a consequence of the adoption of the joint custody law, exposing them to a higher risk of poverty. The paper discusses a possible explanation for these findings, namely that the higher child support payment the mother receives from the non-custodial father in case of joint custody might discourage her from looking for high paid jobs or investing in her career.

Keywords Joint custody law · Bargaining · Difference in difference

JEL Classification J12 · J13

# **1** Introduction

In the last quarter century, the US has witnessed remarkable changes in the family. The social institution of marriage has been increasingly substituted by cohabitation,

Responsible editor: Alessandro Cigno

Daniela Vuri daniela.vuri@uniroma2.it

<sup>&</sup>lt;sup>1</sup> Department of Economics and Finance, University of Rome "Tor Vergata", CEIS, CESIfo and IZA, Via Columbia 2, 00133 Roma Italy

while divorce rates had already risen sharply from the mid-1960s to the early 1980s (Stevenson and Wolfers 2006). Although many scholars have looked at the consequences of the move from mutual consent to unilateral divorce law (Wolfers 2006; Friedberg 1998; Stevenson 2007; Gruber 2004, among others), only recently has the economic literature started to pay attention to another important aspect of divorce, namely the custody of children after divorce.

The allocation of custody is a crucial aspect of any divorce process since it determines the post-divorce living arrangements and potentially has long-lasting consequences on parents and children's well-being. California was one of the first states in the US to introduce a law favoring joint custody at the end of the 1970s and since then it has been extended to most states (Brinig and Buckley 1998). Rather than awarding sole custody to one parent (traditionally the mother) and limited visiting rights to the other (the father), joint custody is intended to preserve the parental role and status of both parents after separation, enabling continued involvement and ensuring the best interest of the child. Under joint custody, the child is expected to spend time with each parent (joint physical custody) and parents need to agree on important decisions regarding their children (joint legal custody). The financial arrangements of parents also change since they are directly called to contribute child expenses.

This study investigates whether joint custody reform affects labor market outcomes of mothers by examining the reform's impact on poverty, earnings, other income received, probability of relying on welfare, and of being in the labor force. The most obvious way custody reform affects maternal labor market outcomes is through the divorce propensity, but the sign of this effect is unclear a priori. For example, married mothers may be less likely to file for divorce because they expect to see their time with their children reduced, with a negative effect on divorce rate. Conversely, joint custody might lead mothers to invest less in their children and marriage due to a lower expected return in these specific marital investments, generating a positive effect on divorce rates. Despite the ambiguous prediction on divorce, custody reform can also directly affect the labor market outcomes of married mothers by altering the balance of power between spouses within marriage (Chiappori et al. 2002; Allen and Brinig 2011; Chiappori et al. 2002; Nunley and Seals 2011). Joint custody may also affect the behavior of divorced/separated mothers, although with potentially opposite effects. On the one hand, there might be a redistribution of childcare responsibilities among ex-partners since separated partners spend less time on parenting and more time in the labor market. On the other hand, divorced/separated mothers might devote more time to the re-marriage market, thus increasing their likelihood of re-marriage and decreasing the labor force participation.

In order to investigate the possible effect of joint-custody law on mothers' labor market outcomes, I use the exogenous variation provided by the custody reform across the US states as a natural experiment to identify the average treatment effect of the reform. The same identification strategy to study the impact of joint custody on several parental and children outcomes in the US has been used by Halla (2013), Teng (2008), Nunley and Seals (2011), Altindag et al. (2015), and Allen, Nunley and Seals

(2011). Mothers with children who live in states that introduced the joint custody law between 1970 and 2000 constitute the treatment group, while the comparison group is made of mothers who live in states that had yet to adopt joint custody law by 2000. This study provides two important extensions to the existing literature. First, it estimates the effect of the joint custody reform on several economic outcomes of mothers and not only on the labor force participation and hours of work of (married) mothers (see Nunley and Seals 2011; Altindag et al. 2015; Halla 2013). Second, the results for single mothers are interpreted in the light of a new mechanism, the increase in child support payments, that, to the best of knowledge, has yet to be linked to mothers' behavior. Allen et al. (2011) have investigated the effect of joint custody legislation on the child support received by single mothers; Halla (2013), Böheim et al. (2016), and Nunley and Seals (2011) have studied the effect of joint custody reform on the labor force participation of women and married mothers respectively, but none of these studies has linked these three aspects together.

The findings indicate that only divorced/separated/never married (single, henceforth) mothers are negatively affected by the joint custody law in terms of a decrease in earnings, which exposes them to a higher risk of poverty. The probability of being in the labor force is instead unchanged with the passage of the custody law. Importantly, the results are not driven by changes in the composition of family structure as a consequence of the introduction of the joint custody law: on aggregate, the proportion of children living with a single parent or in intact families does not change. This is extremely relevant for the analysis because it rules out that selection effects due to a changing composition in the stock of married and unmarried population might affect the estimates and allows us to focus exclusively on single mothers, for whom the joint custody reform turns out to have an impact. The results are robust to controlling for unilateral divorce, specific property division laws and child support legislation. No evidence is found that differences in the exposure to the reform matter for mothers' labor market outcomes. Child's age at the reform does not matter, except for the odds of living in poverty, which increases as the child ages, and for the probability of receiving "other income," which is higher when the child is young.

The findings can be explained by the higher child support payments a mother receives from the non-custodial father in case of joint custody, which might discourage her from looking for a high paid job or investing in her career. Since there is no effect on mother's labor force participation, it can be argued that the drop in earnings might be due to a reduction in the number of hours worked. Moreover, since the probability of living in poverty increases, the increase in compliance with child support payment seems not to be enough to compensate the mother's reduction in earnings - especially for low income mothers.

The rest of the paper is organized as follows: Section 2 describes the evolution of the law concerning the custody decision over time; Section 3 briefly discusses the theoretical framework and related work on this topic; Section 4 describes the data employed in the analysis and the sample selection; Section 5 illustrates the identification and empirical strategy; Section 6 presents the estimation results and some robustness checks; while Section 7 provides possible interpretations of the results.

# 2 Legal background

Looking at the custody decision making in historical context, the determination of which parent should have custody of children in a divorce proceeding has gone through several changes in the US (Brinig and Buckley 1998). In case of divorce, until the mid-nineteenth century, fathers had a near absolute right to custody. In the late 1800s, following the increasing specialization within the family with the father acting as the breadwinner and the mother looking after the children and the house, the paternal preference was gradually replaced by the maternal preference. In many states, maternal preference took the form of the "tender years doctrine" (under age 6), that presumes the mother is the more suitable custodian for children in case of parental separation.<sup>1</sup>

By the 1920s, the maternal preference for custody, regardless of the child's age, became firmly fixed and was encoded in the statutes of all 48 states. The maternal presumption for custody remained in this form for many decades, to be challenged only after the divorce rate began to rise in the 1960s. Legal practice changed between the 1970s and the 1990s, after the Family Court of New York invalidated the "tender years doctrine" in New York State (*Watts v. Watts*, 1973) in favor of the doctrine of *the best interest of the child* (BIOC), which makes no reference to the gender of the gender-neutral custody assignment (Mason 1999). Despite this legal change, many courts adopted the "primary caretaker presumption," which operated in a gender biased manner since in the vast majority of the cases the primary caretaker was identified as the mother (Cancian and Meyer 1998; Rose and Wong 2014).<sup>2</sup>

Many other changes occurred before the mid-1970s, i.e., women's increasing labor force participation, the feminist movement, fathers' claims of sex discrimination in custody decisions, and constitutional concerns for equal protection (Jacob 1988).

All these changes prepared the path for the next step, the joint custody after divorce. Between the late 1970s and the early 1980s, the joint custody statute was enacted in California, Connecticut, Kansas, Kentucky, and Oregon. By 1990, more than 40 states had shared parenting statuses in which joint custody was either an option or preference, and most other states had recognized the concept of joint custody in case law. Contrary to other family-law reforms, opinions from experts were rare in the legislative discourses on joint custody (Jacob 1988). In the absence of scientific evidence on the consequences of joint custody for children, the reform was passed in relative obscurity (Jacob 1988).

<sup>&</sup>lt;sup>1</sup>In the "tender years doctrine," children were no longer viewed as economic assets or as workers under the control of their fathers; instead they came to be seen as individuals with their own needs and in the case of young children, they were recognized to have special needs for nurture, normally satisfied by the mother. Current laws do not give special attention to child nurture and do not recognize the importance of developmental stages. In other words, two-year old children are treated the same as children of twelve.

<sup>&</sup>lt;sup>2</sup>Utah is a typical example of inconsistency between state statutes and case law. Although the explicit maternal presumption was repealed by the legislatures in 1977, the courts continued to practice maternal preference toward mothers in child custody cases until 1986; the judicial preference in favor of the mother was explicitly abolished in the case *Pusey v. Pusey* in 1986 (Chen 2013).

Nowadays, in most US states, the most common arrangement is joint legal custody and sole or primary physical custody to one of the parents.<sup>3</sup> Law documents say that in a typical divorce involving at least one child, permanent physical custody is awarded to the parent with whom the child will live most of the time. Usually, the custodial parent shares joint legal custody with the noncustodial parent. As a matter of fact, the introduction of the joint custody law has not changed the living situation of the families but only the legal aspects of the custody arrangements. Further evidence of this is provided in Section 6.1.

There is evidence that judges have followed the prescriptions of the new custody laws. Halla (2013), using data from divorce certificates provided by the *National Vital Statistics System* of the *National Center for Health Statistics*, shows that the probability of a joint custody award increases by 3 percentage points each year after the introduction of the joint custody. Further support for this finding is provided by Allen and Brinig (2005), Brown et al. (1997), and Maccoby and Mnookin (1997) who suggest that there was a substantial increase in the number of joint custody arrangements after the reform.<sup>4</sup> Evidence on the application of the law is particularly important in the identification strategy (see Section 5) because it proves that the reform did not exist only on papers but it was actually implemented by states.

## **3** Theoretical considerations

Changes in child custody law may influence the bargaining that takes place within marriage and modify the resource allocation in it. *Exit threat* bargaining models (Manser and Brown 1980; McElroy and Horney 1981) emphasize the role played by each spouse's best outside alternative to marriage as relevant in determining the household distribution. Any policy changes, like custody reform, that alter divorce-threat point potentially affect allocation decisions. In this case, the introduction of joint custody will, ceteris paribus, improve the bargaining position of the spouse who most likely would have been the non-custodial parent, i.e., the father, while the opposite holds true for the mother. By contrast, in collective models, allocation decisions are the result of bargaining over possible decisions and do not assume any specific decision-making process (Browning and Chiappori 1998; Chiappori et al. 2002).

<sup>&</sup>lt;sup>3</sup>Nearly all the adopted states have distinguished in their legislation between legal and physical custody. Legal custody refers to the parental right to make major decisions regarding child's health, education, welfare etc. Physical custody refers to the living arrangement of the child on a day-to-day basis. In states permitting or encouraging joint legal and physical custody arrangements, it is difficult to determine what percentage of parents have joint legal or physical custody, as the data must be obtained from individual divorce decrees. For this reason no distinction is made between the two in this analysis and the child custody reform that is examined is meant to capture the overall change to both types of joint-custody arrangements.

<sup>&</sup>lt;sup>4</sup>Allen and Brinig (2011) found that joint parenting in Oregon led to changed custody arrangements, but it came through changes in sole and split custody (some of the children living with one parent, and some of the children living with the other parent) and not in joint custody. However, these results should be interpreted with caution since they only refer to one state and are based on a before/after approach.

Most of the theoretical literature has focused on how changes in family laws predict changes in labor force participation of married couples. Both divorce-threatbargaining and collective models provide clear predictions concerning the effect of family-law reforms on household labor supply, with men allocating more time to household work and less to market work, and the opposite for women (Altindag et al. 2015). This could have a twofold advantage: raise the mother's bargaining position within marriage by increasing her labor market skills and insure her against increased divorce-threat of fathers who, after the custody reform, can rely on higher expectations of shared child custody. Similarly, Grossbard-Shechtman (1995) shows that laws affecting divorce settlements, like the no-fault divorce, can change the equilibrium outcomes in the marriage market by reducing the demand for spousal labor by men and increasing women's labor market participation. This reasoning can be easily extended to the custody reform. This suggests that bargaining power and labor supply are inversely related. In contrast, Gray (1998) shows that married women reduce their labor supply when they lose bargaining power.

However, Lundberg and Pollak (1993) point out that most of the bargaining literature focuses on intra-household bargaining, rather than on couples who have already separated or are close to dissolving their marriage. For an average married couple, far from the possibility of divorce, changes in legal rules regulating child custody might have a weak impact on individual behavior. On the contrary, (close to) separated couples might be the most sensitive to changes in legal rules that immediately influence their bargaining position; for them, changes in the allocation of property and custody settlements decided by courts at divorce might be especially relevant. Joint custody reform might therefore affect labor market outcomes of single mothers through different channels than the ones of married mothers, with no clear-cut predictions. On one side, under the new custody regime single mothers might spend on average less time on parenting, thus increasing the time devoted to work. On the other side, they might increase the time spent on the re-marriage market, which in turn increases their probability of re-marriage and reduces their labor market participation.

Ascertaining how married and single mothers respond to the joint custody reform is ultimately an empirical question upon which this paper intends to shed some light.

## 3.1 Related work

In the last decades, there has been an ongoing debate among the supporters and opponents of custody reform. Proponents typically argue that children may benefit from more parental resources, in particular from fathers both before and after divorce, and the benefits might extend to parents as well. According to the *bonding theories*, before divorce a father might be more willing to invest in, or bond more closely with, his children (and possibly with his wife) if he is not going to lose access to his child(ren) in case of divorce (Brinig and Buckley 1998). After divorce, according to the *monitoring theories*, on a move to joint custody a father might be more ready to give child-support payments if joint custody solves the agency-cost problem arising from the fact that the non-custodial parent (usually the father) is not able to monitor how the child-support payments are actually spent (Brinig and Buckley 1998; Weiss and Willis 1985). Divorced parents who share physical custody of their children also share the time and energy costs of homemaking and parenting, facilitating labor force participation or investment in their careers (Morrow 1991).

Opponents emphasize the countervailing effects of joint custody on families. The shift from the maternal custody regime to the joint custody regime might disadvantage women in their property negotiations if they give more value to sole custody than the spouse gives to joint custody and trade-off property for sole custody (Brinig and Buckley 1998; Rasul 2006). Moreover, joint custody might not work well in high-conflict families (Mason 1999) and eventually might lead to less support for children if the court establishes a reduction in child support payments entitled to a mother (Morrow 1991; Seltzer and Maralani 2001; Mason 1999). Finally, mothers' careers and educational opportunities might be restricted by limitation in their geographical mobility, incorporating a gender-based discrimination that reinforces the economic dependency of women (Bartlett and Stack 1991).

A number of researchers have attempted to provide empirical evidence of the effects of joint custody reform on individual behavior and children's well-being. Research on the relationship between changes in family law and individual behavior has mainly focused on the effect of custody law on divorce, marriage, fertility and employment decisions in the US. Brinig and Buckley (1998) find a negative effect of the introduction of joint custody law on divorce rates. Recently, Halla (2013) has shown that the introduction of joint custody led to a long-run increase in marriage rates and fertility, a decrease in female labor market participation, and a not significant effect on divorce rates probably due to countervailing forces, i.e., costs effects and behavioral effects, which move in opposite directions.<sup>5</sup> Further, he finds that male suicide rates and domestic violence fell in treated states. Qualitatively similar effects are found for marriage and fertility rates in Austria and Germany, but differently from the previous study, the divorce rate decreased after the custody reform (Böheim et al. 2012). Focusing only on married mothers, Nunley and Seals (2011) find an increase in labor force participation in law-adopting states in the US. Similarly, Altindag et al. (2015) show that custody reform encouraged a reallocation of time within marriage, with married mothers increasing the time devoted to work in the market and married fathers increasing the time allocated to work at home. $^{6,7}$ 

Several studies have looked at the effect of custody arrangement on children's well-being after divorce with mixed results. Consistent with the *bonding theories*, past research has shown that fathers awarded joint custody spend more time with their children (Seltzer 1998) with positive effects on children's behavioral and emotional

<sup>&</sup>lt;sup>5</sup>Possible reasons for the difference in findings between the two studies could be attributed to differences in the definition of divorce rates, in the time window and in the functional form of their estimation models. <sup>6</sup>The differences between Halla (2013) and Nunley and Seals (2011) and Altindag et al. (2015) might be explained by the different sample selection and estimation procedure.

<sup>&</sup>lt;sup>7</sup>Recently, two studies have examined the effect of the introduction of the gender-neutrality in the US, according to which fathers have equal rights to custody at divorce. Chen (2013) shows that gender-neutral laws increases divorce rates in the long run, while Rose and Wong (2014) finds that the gender-neutral law decreases the hazard into marriage. Rose and Wong (2014) also investigate the subsequent move toward marriage neutrality, which extends to unmarried fathers the possibility to gain sole custody if they qualify as a suitable custodian. They find no evidence of a reform effect on marriage rates.

adjustment (Bauserman 2002). Joint custody is found to increase the involvement of nonresident parents in their children's lives, although this effect diminishes over time (Huang et al. 2003). Teng (2008) shows that the educational attainment of children of divorced or separated parents increases in states adopting joint custody law. Other studies have however highlighted the negative effect of joint custody law on (i) children exposed to parental conflict, especially among families who showed a high intensity of inter-parental hostility at an earlier time (Maccoby et al. 1991; Kuehl 1989; Mason 1999), (ii) on children of intact families that experience a decrease in their educational attainment (Teng 2008), and (iii) on children raised in a joint custody regime who reduce their probability of attending a private school (Nunley and Seals 2011) and experience worse labor market outcomes (Maiti 2015).

A few studies have focused on the impact of child-custody on the receipt and level of child-support income, with no clear-cut indications. Confirming the predictions of the *monitoring theories*, some studies find that fathers awarded joint custody are more willing to comply with child support orders than non custodial fathers (Del Boca and Ribero 1998; Huang et al. 2003; Allen et al. 2011). However, other studies find no effect (Seltzer and Maralani 2001; Seltzer 1998; De Blasio and Vuri 2013; Allen and Brinig 2011), or evidence of heterogeneity according to marital status (Allen et al. 2011). In particular, joint custody reform has an effect only on divorced mothers, while separated and never-married mothers are unaffected by the reform.

The present analysis improves upon much of the previous research by providing a broader view of the labor market consequences of joint custody reform for single mothers and by linking their market behavior with the finding of an increase in child support recipiency.

## 4 The data and the sample selection

The primary data for the 50 US states and the District of Columbia come from the 1960 to 2000 Census Five-Percent Public Use Micro Sample (IPUMS).<sup>8</sup> The focus here is on mothers between 19 and 55 years of age whose youngest child is under 18 years and therefore still potentially involved in child custody disputes.<sup>9</sup>

Several variables are used to measure mothers' labor market outcomes: (i) poverty, (ii) earnings, (iii) other income and its dummy, (iv) labor force participation, and (v) being "on welfare." Poverty is an indicator equal to one if the mother falls below the official "poverty line"; the variable "earnings" reports the mother's total pre-tax wage and salary income, that is, money received as an employee for the previous year; "other income" is computed as an individual's total income, or losses, from sources

<sup>&</sup>lt;sup>8</sup>The 1960 Census sample is included to allow for a credible identification of *preexisting* state specific trends (Gruber 2004; Wolfers 2006).

<sup>&</sup>lt;sup>9</sup>Stepmothers are excluded since the census questions identifying this group changed in 1990. Moreover, remarried families have many confounding effects that make the impact of joint custody less clear-cut. Widowed mothers—who represent a very small fraction of the population (only 0.2%) mainly due to the selection of relatively young women in the sample—are excluded. As expected, the inclusion of this group in the sample does not change the results.

not included in the other IPUMS person-record income variables;<sup>10</sup> a dummy variable "being on welfare," equal to one if the mother receives income from any public assistance programs and zero otherwise.<sup>11</sup> Finally, the set of income-related variables is completed with "labor force participation (LFP)," a dummy variable equal to one when the mother is in the labor force.<sup>12</sup> All the income variables are expressed in 2000 dollars. Since some income variables have different top codes over years, consistent values of the top codes are generated by imposing the real values of the 1960s top codes (the most restrictive) in each year, as suggested by Genadek et al. (2007). This procedure makes the variables used in the analysis comparable across years. Summary statistics for the whole sample and separately for married mothers and single mothers are presented in Table 1. Panel A reports the summary statistics for the outcomes, Panel B for the control variables (age, level of education, and race of the mother, total number of children and age, gender and year of birth of the youngest child in the household).

During the period of analysis, many states have changed their child custody law, as shown in column 1 of Appendix Table 8. The coding of the custody law for each state is from Teng (2008), who distinguishes states with custody law awarding joint custody only if parents are in agreement from states which allow application by only one parent.<sup>13</sup> This distinction is important because if agreement is necessary, the threat of opposing is credible and may affect intra-household bargaining power. For six states (Colorado, Nebraska, North Carolina, Oregon, Vermont, and Wisconsin) both the year of the more restrictive custody rule and the year of the more permissive one are reported.<sup>14</sup>

Appendix Table 8 also provides a summary of the rules in place over the period of analysis by state. The unilateral divorce coding in column 2 is from Gruber (2004), while the data on property division types (columns 3 and 4) are from Gray (1998) and Stevenson (2007). Columns 5 to 12 report the year of introduction of eight child support laws which measure the vigor of state child support legislation (from Huang 2002 and Huang et al. 2002). The legislative activity covers two steps of the enforcement process: obtaining adequate child support orders and collecting child support

<sup>&</sup>lt;sup>10</sup>Results related to the "other income" variable should be interpreted with caution since this variable is not perfectly comparable across years. As reported in the IPUMS documentation (see Ruggles and Sobek (1997)), in 1960 respondents were simply asked to declare the amount of money income received from sources other than wages and self-employment. Beginning in 1970, other possible sources of additional income, such as the sale of personal property, lump-sum insurance and inheritance payments, payments "in kind", veterans' payments, unemployment compensation, child support, and alimony were added to the list of excludable variables. The exclusion of the year 1960 does not change the results.

<sup>&</sup>lt;sup>11</sup>The variable "being on welfare" includes federal/state Supplemental Security Income (SSI) payments to elderly (age 65+), blind, or disabled persons with low incomes, Aid to Families with Dependent Children (AFDC) and General Assistance (GA).

<sup>&</sup>lt;sup>12</sup>Unfortunately, it was not possible to use the number of hours worked because this variable is not collected throughout the Censuses.

<sup>&</sup>lt;sup>13</sup>Brinig and Buckley (1998) provided another code for joint custody but it is update only up to 1993. For this reason this paper prefers the coding proposed by Teng (2008), who has nevertheless proved his results to be robust to the use of the Brinig and Buckley's coding.

<sup>&</sup>lt;sup>14</sup>The analysis uses the more restrictive coding but the results are unchanged if the other rule is considered. Results are available on request.

	All samp	le	Single n	nothers	Married	mothers
	Mean (1)	Std. Dev. (2)	Mean (3)	Std. Dev. (4)	Mean (5)	Std. Dev (6)
Panel A: Outcomes						
Poverty	0.134	(0.341)	0.381	(0.486)	0.088	(0.283)
Log earnings	9.33	(11.93)	9.48	(10.91)	9.29	(12.15)
Log other income	8.08	(11.93)	7.96	(14.05)	8.11	(11.25)
Other income (0/1)	0.258	(0.438)	0.357	(0.479)	0.240	(0.427)
On welfare	0.054	(0.227)	0.224	(0.417)	0.017	(0.130)
LFP	0.550	(0.497	0.732	(0.443)	0.516	(0.499)
Panel B: Controls						
Mother's variables						
Married	0.842	(0.365)	_	-	_	_
Divorced	0.078	(0.268)	0.492	(0.500)	_	_
Separated	0.035	(0.185)	0.224	(0.417)	_	_
Never married	0.045	(0.207)	0.284	(0.451)	_	_
Age	35.46	(8.12)	34.08	(8.16)	35.72	(8.08)
Less than high school	0.238	(0.426)	0.259	(0.438)	0.234	(0.424)
High school	0.401	(0.490)	0.391	(0.488)	0.403	(0.490)
College graduate	0.361	(0.480)	0.350	(0.477)	0.363	(0.481)
White	0.836	(0.370)	0.624	(0.484)	0.876	(0.329)
Black	0.110	(0.313)	0.307	(0.461)	0.073	(0.261)
Other race	0.053	(0.224)	0.069	(0.253)	0.051	(0.218)
Hispanic origin	0.074	(0.262)	0.099	(0.299)	0.069	(0.254)
Tot n. of children	2.08	(1.18)	1.85	(1.10)	2.12	(1.19)
Youngest child's variables						
Female	0.490	(0.499)	0.497	(0.499)	0.489	(0.499)
Age	7.00	(5.43)	7.65	(5.28)	6.88	(5.45)
Ν	1,441,684	4	228,089		1,213,59	95

Table 1 Sample means for mothers with children under 18 years and by marital status

Source. The 1960, 1970, 1980, 1990, and 2000 Integrated Public Use Microdata Series (IPUMS).

Notes: Log earnings is computed on 842,615 observations for the whole sample, on 170,135 observations for the sample of single mothers and on 672,480 observations for the sample of married mothers; log of other income is computed on 372,553 observations for the whole sample, on 81,340 observations for the sample of single mothers and on 291,213 observations for the sample of married mothers; on welfare is not available for year 1960 and it is computed on 1,194,473 observations for the whole sample, on 215,170 observations for the sample of single mothers and on 979,303 observations for the sample of married mothers.

payments. The eight child support laws considered are paternity and genetic tests, numerical guidelines, presumptive guidelines, wage withholding under delinquency, immediate wage withholding for a new case, universal wage withholding, and state income tax interception. Numerical guidelines advise judges concerning child support laws and are non-binding. Presumptive guidelines require that these numerical guidelines are used unless the judge can cite good reasons to deviate from them. The wage withholding under delinquency measure indicates that the state has developed a system, similar to income tax withholding, that allows the deduction of child support obligations and any arrearages accrued from the obligor's pay-check. Immediate withholding laws require all new or modified support orders for welfare recipients to have immediate withholding of support. Finally, the state income tax interception measure indicates that the state has procedures available to garnish state income tax refunds, up to the amount of overdue child support. A 2-year lag between legislative enactment and implementation is assumed.

## 5 Identification and estimation strategy

As shown in Appendix Table 8, states vary substantially in the year they enacted joint custody law and there is convincing evidence that the introduction of joint custody can be considered random. For instance, Halla (2013) has shown that the introduction of joint custody does not follow any systematic geographical patterns, there is no relationship between the size of the state and the timing of the reform and there are no links with political ideology. A similar conclusion has been reached by Altindag et al. (2015), who show that trends in married mothers' and fathers' time allocated to market work and household work in early-adopting states are similar to those in late-adopting states. Moreover, they show that labor market participation, weeks worked, and hours of work of married mothers follow similar patterns in states that will adopt and will not adopt the joint custody in the years preceding its introduction. This apparent randomness in the introduction of joint custody law across states and time is also supported by historical accounts. Jacob (1988, ch.8) reports that joint-custody reform was discussed by a small group of proponents and passed legislatures in relative obscurity.

Taking advantage of the randomness in the introduction of joint custody law and the evidence on the change in joint custody arrangements following custody reform (mentioned in Section 2), the impact of the reform on mothers' labor market outcomes can be estimated using a standard Dif-in-Dif approach. The different timing in the introduction of joint custody across US states provides a useful quasiexperimental setting to study the causal effect of joint custody on mothers' outcomes. Identification is thus obtained through variation across states and over years in which states implemented the joint custody reform.<sup>15</sup> Mothers with dependent children who live in states that change their laws to favor joint custody by year 2000 constitute

<sup>&</sup>lt;sup>15</sup>This approach has been used extensively in the literature investigating the consequences of introducing the unilateral divorce law (see Gruber 2004; Friedberg 1998; Stevenson 2007 and 2008; Stevenson and Wolfers 2006, Wolfers 2006, among others). The same approach has been used to study the consequences of adopting the joint custody law on several outcomes (see Halla 2013; Teng 2008; Böheim et al. 2012 and 2016; Altindag et al. 2015; Nunley and Seals 2011; Allen et al. 2011; Rose and Wong 2014; Chen 2013). For a general overview of the difference-in-difference models see Angrist and Pischke (2009) and Wooldridge (2003).

the treatment group, while those who live in states that did not institute joint-custody reform before 2000 represent the control group.

In the first part of the analysis, I estimate the impact of legal changes on mothers' outcomes in the same time period, i.e., the *contemporaneous* effect of legal changes. This approach is similar to Friedberg (1998) and Wolfers (2006). Specifically, I estimate a Dif-in-Dif model of the form:

$$y_{ist} = \beta \times JC_{st} + \mu_s + \tau_t + \mu_s \times t_t + \gamma X_{ist} + \varepsilon_{ist}$$
(1)

where  $y_{ist}$  represents the outcome of interest for mother *i* resident in state *s* and observed in year *t*. The variable  $JC_{st}$  is dummy variable equals to one if the law is in place in state *s* at year *t* and zero otherwise. All estimates include a vector of state dummies  $\mu_s$  and Census year dummies  $\tau_t$ .

The focus of the analysis is on the parameter  $\beta$ , which captures the difference in the outcome variables between mothers in joint-custody versus sole-custody states.  $X_{ist}$  represents a vector of mother characteristics: age, race, ethnicity (a dummy variable for Hispanic origin), family structure indicator variables, number of children at home, educational attainment indicator variables (high school graduate, some college and college). Some child characteristics are also included: sex, age, and year of birth of the youngest child in the household, i.e., the one for whom the custody agreements are more binding. Census year dummies  $\tau_t$  control for mothers' outcomes growth common to all states; state fixed effects  $\mu_s$  control for unobserved influences on mothers' outcomes that vary across states. To take into account the fact that some factors influencing individual outcomes may vary within a state over time, statespecific linear time trends are included, indicated by the interaction  $\mu_s \times t$  in Eq. 1, which capture the influence of any omitted time-varying state-level variables.<sup>16</sup> All the regressions include a full set of child age by year, by state of residence and by state of birth interactions to allow for differential time or state patterns by age of the child. Finally,  $\varepsilon_{ist}$  captures idiosyncratic variation by individual. Recent analyses demonstrate that pervasive serial correlation in state level difference-in-difference models may produce downward biased standard errors (Bertrand, Duflo and Mullainathan, 2004). For this reason, Huber-White SEs clustered at state level are used throughout, which are robust to arbitrary forms of error correlation within a state.

Ideally, the relevant variables would be measured at the year of divorce when joint custody is usually decided instead of the survey year; unfortunately, census data do not provide this information. However, given that the custody arrangement is decided according to the prevalent law in the individual state of residence, it can be assumed that mothers make their decisions based on the prevailing custody law in their current

<sup>&</sup>lt;sup>16</sup>As pointed out by Nunley and Seals (2011) for example, the increase in female labor force participation rate and fathers' involvement in childrearing could be correlated with the passage of joint-custody reform. Failure to account for these state-level covariates could bias the estimates. However, fixed trend models are quite demanding when using census data because there are few time-series observations (in this case only five). Moreover, trends measuring decade time windows will not be able to capture the short-run endogeneity, but only the long-term one (Gruber 2004; Angrist and Pischke 2009).

state of residence (Halla 2013). The issue of endogeneity in moving will be explored in Section 6.4.

It is important to mention that year, state, state-specific linear trends, and interaction effects capture the differences in mother's outcomes across states and their changes *only* under the hypothesis that there are no selection effects into different family structures, like marriage or divorce. In absence of selection, the coefficient on  $JC_{st}$  identifies the net effect of joint custody, otherwise it only captures a gross effect (Teng 2008). Evidence on the absence of selection effects will be provided in Section 6.1.

The *contemporaneous* DID estimator used in Eq. 1 imposes the assumption of immediate and constant response to the reform. However, labor market outcomes might respond to the adoption of the custody law via a dynamic process. To take into account this possibility, this paper uses the empirical strategy suggested by Wolfers (2006) and adopted by Halla (2013) and Chen (2013) who introduce a dynamic structure of the policy intervention through dummy variables for the years after the adoption of the joint custody law. Specifically:

$$y_{ist} = \sum_{r=0}^{m} \beta_{-r} J C_{s,t-r} + \mu_s + \tau_t + \mu_s \times t_t + \gamma X_{ist} + \varepsilon_{i,s,t}$$
(2)

where  $JC_{s,t-r}$  denotes a series of binary variables equal to one if a state *s* has introduced joint custody *r* years ago. Lags up to 15 years after the reform are included. This flexible specification allows us to trace out the outcomes' full adjustment path.

The validity of the DID approach is based on the common trends assumption which requires that in the period before the treatment the trend in the outcome(s) in the treated and control groups is the same or very similar (Angrist and Pischke 2009). This is equivalent to saying that the unobserved determinants of the mothers' outcomes for the treated and controls would have followed parallel trends had there been no exposure to the treatment. While the common trend assumption is no testable by assumption, it is possible to assess its plausibility using the "Granger Causality" approach taken in Autor (2003) and described in Angrist and Pischke (2009). The Granger causality testing means checking whether, conditional on state and year effects, past policy variable  $JC_{st}$  predicts  $y_{ist}$  while future  $JC_{st}$  does not. Therefore, a model of the following form is estimated:

$$y_{ist} = \sum_{r=0}^{m} \beta_{-r} J C_{s,t-r} + \sum_{r=1}^{q} \beta_{+r} J C_{s,t+r} + \mu_s + \tau_t + \mu_s \times t_t + \gamma X_{ist} + \varepsilon_{i,s,t}$$
(3)

which allows for q leads  $(r_{+1}, r_{+2}, r_{+q})$  or anticipatory effects and m lags  $(r_0, r_{-1}, r_{-m})$ . In absence of anticipatory effects, the coefficients for future policy changes should be zero and this could be interpreted as evidence of absence of a treatment effect before the treatment. Results from estimation of Eqs. 1, 2 and 3 are reported in Section 6.

# 6 Impact of joint custody laws on mother's outcomes

# 6.1 Compositional changes in family structures

Before turning to the estimates of the effect of joint custody reform on mother's labor market outcomes, it is important to rule out possible selection effects into and out of marriage and fertility that might bias our estimates. Recently, Halla (2013) has shown that the introduction of joint custody in the US led to a long-run increase in marriage rates but had no effect on divorce rates, which he attributes to countervailing forces. Looking at children aged between 15 and 18 years, Teng (2008) shows that the adoption of unilateral divorce and joint custody laws do not change the probability for a child of living in an intact family or in a single parent family. Finally, Böheim et al. (2016) find that the introduction of a custody reform in Austria has a negative effect on marital birth rates but it vanishes when restricting the sample to more comparable districts.

To assess the impact of joint custody regulation on family structure in my sample, regressions of joint custody on marital status of parents of children below 18 years of age are run, as in Gruber (2004). Specifically, the sample is made of all children aged 0–18 living with a least one biological parent (stepparents have been excluded) and the dependent variables are whether the child's mother or father are divorced, separated, never married, married, or widowed. As Gruber (2004) has proved that making divorce easier increases the probability that children are living with a divorced parent, a dummy for adopting unilateral divorce law is included. As additional outcome, the number of siblings the child is living with is considered as a proxy for family size (and therefore of fertility). The regression performed is as follows:

$$f_{ast} = \alpha + \beta_1 J C_{ast} + \beta_2 U D_{ast} + \beta_3 Race_{ast} + \mu_s + \mu_s \times t_t + \tau_t + \lambda_a + \lambda_a$$
$$\times \tau_t + \mu_s \times t_t + \varepsilon_{ast} \tag{4}$$

where *a* indexes ages, *s* states, *t* years;  $f_{ast}$  is the cell mean divorce rates (or separated, married, never married, and widow rates) of the mother/father the child is living with, or the number of siblings. *JC* is the dummy for joint custody laws as defined in Section 4; *UD* is a dummy for unilateral divorce law in state *s* in year *t*. Race are for % Blacks and % White, respectively in the cell;  $\mu_s$ ,  $\tau_t$  and  $\lambda_a$  are full set of dummies for state, year, and age group;  $\lambda_a \times \tau_t$  is a full set of age\*year interactions to control for differential time patterns by age. As in Gruber (2004), this model controls for fixed factors that vary by age, state, or year and it is identified by the passage of joint custody and divorce laws over time. I consider the effect of the two legal reforms on the mother/father's marital status. Three different regressions are performed: (i) the first including only unilateral divorce law control to replicate results in Gruber (2004) and therefore to check the validity of our estimates;<sup>17</sup> (ii) the second including only joint custody law control; (iii) the last including both divorce

<sup>&</sup>lt;sup>17</sup>Unilateral divorce, which allowed divorce with the consent of just one rather than both spouses, was rare before the late 1960s, but it was in place in most states by the mid-1970s. Of the 50 states, in 2006, five had yet to adopt any form of unilateral divorce.

and joint custody law controls. State-specific trends are controlled for and standard errors are corrected for clustering on state (Bertrand et al. 2004).

As a preliminary step, it is important to establish whether the changes in child custody law and the movement to unilateral divorce law are correlated with each other. If the two movements are jointly determined, disentangling the two effects would be difficult. As proposed by Chen (2013), only states that have completed both legal reforms are considered and the timing of the unilateral divorce law is then regressed on the timing of the child custody law. The correlation between the times of changes in the two laws turns out to be negligible (coeff. 0.0394, s.e. 0.467), suggesting that the adoption of unilateral divorce law does not predict when a state changed its custody law. The independence of the two legal trends allow to estimate the effect of child custody law controlling for the adoption of unilateral divorce in the same regression.

The results in Table 2 (columns 1 and 4) are in line with Gruber (2004). There is evidence of a positive effect of the adoption of unilateral divorce on the proportion of children living with a divorced mother/father (Panel A - columns 1 and 4), while the effect is negative (positive) for the odds of living with a separated mother (father) (Panel B - columns 1 and 4). On the contrary, the adoption of the joint custody law does not affect the probability that a child lives with his/her divorced mother/father (Panel A - columns 2 and 5). There is only some evidence that the adoption of the joint custody law increases the proportion of children living with a separated mother (Panel B - column 2), but not with a separated father (Panel B - column 5). Concerning the other marital status variables, both unilateral divorce and joint custody laws have hardly any effect on the odds of a child of living with a married, never married or widowed parent. Only the odds for a child of living with a married father decreases after the adoption of the unilateral divorce law even after controlling for the introduction of the joint custody law (column 6). No effects are found for the number of siblings in the family the child is living with. Overall, it does not seem to be any significant effect of the adoption of joint custody law on family composition, once unilateral divorce is controlled for (columns 3 and 6), with the exception of living with a separated mother (Panel B, column 3).<sup>18</sup>

Nunley and Seals (2011) and Altindag et al. (2015) propose a different method to address potential selection issues generated by custody reform based on the inverse probability procedure (IPW) suggested by Wooldridge (2002). This procedure gives less weight to observations who are more likely to be in the sample because of custody reform. In their case, the weights are obtained through a two-step procedure. In

<sup>&</sup>lt;sup>18</sup>Altindag et al. (2015) use a different strategy to control for possible selection effects. They examine a sample of married couples who wed before 1977, i.e., before the widespread adoption of joint custody law across states. For those couples, the decision to marry should not be influenced by the passage of joint custody law because it was not in place. The estimates based on the restricted sample do not differ from those on the full sample confirming that selection is not an issue in the custody reform context. A similar approach has been adopted by Böheim et al. (2016) who restrict the sample only to marriages formed before the reform. Halla (2013) restricts the sample to married women and obtain similar results to the whole sample of women, supporting again the hypothesis that selection into marriage is not an issue. Unfortunately, it is impossible to adopt the same strategy in this paper because the year of marriage is not recorded in the IPUMS data.

	Child is liv	ing with mot	her:	Child is livi	ng with fath	er:
	(1)	(2)	(3)	(4)	(5)	(6)
Panel A: Divorced						
Unilat. Divorce	0.008***		0.008***	0.002		0.002
Joint Custody		-0.001	-0.002		-0.000	-0.001
		(0.002)	(0.002)		(0.001)	(0.001)
Panel B: Separated						
Unilat. Divorce	-0.003*		-0.003*	0.001**		0.001**
	(0.002)		(0.002)	(0.000)		(0.000)
Joint Custody		0.004**	0.004**		0.000	-0.000
		(0.002)	(0.002)		(0.001)	(0.000)
Panel C: Never Married						
Unilat. Divorce	-0.001		-0.001	0.001		0.001
	(0.002)		(0.002)	(0.001)		(0.001)
Joint Custody		-0.001	-0.001		0.001	0.000
		(0.001)	(0.001)		(0.001)	(0.001)
Panel D: Married						
Unilat. Divorce	-0.003		-0.003	$-0.005^{**}$		-0.005 **
	(0.004)		(0.004)	(0.002)		(0.002)
Joint Custody		-0.003	-0.003		-0.000	0.001
		(0.003)	(0.003)		(0.001)	(0.001)
Panel E: Widowed						
Unilat. Divorce	-0.001		-0.001	0.001		0.001
	(0.001)		(0.001)	(0.001)		(0.001)
Joint Custody		0.001	0.001		0.000	-0.000
		(0.001)	(0.001)		(0.001)	(0.001)
Panel F: N. of Siblings						
Unilat. Divorce	-0.025		-0.024	-0.036*		-0.036
	(0.019)		(0.019)	(0.020)		(0.023)
Joint Custody		-0.003	-0.003		-0.007	-0.001
		(0.013)	(0.013)		(0.016)	(0.018)

Table 2 Compositional effects of joint custody and unilateral divorce laws

Source: The 1960, 1970, 1980, 1990, and 2000 Integrated Public Use Microdata Series (IPUMS).

Notes: Dependent variables are the cell means at the state by year by age level. Control variables include a full set of state, year and child age dummies, child age by year interactions, % Black, % White. Standard errors are clustered by state of residence. Symbols: \*\*\* significant at 1%; \*\* significant at 5%, \* significant at 10%

the first step, they estimate the probability that a child lives in an intact family using a probit regression on the full sample, controlling for individual and state-level characteristics. In the second step, they estimate the predicted probability of being married. The inverse of the predicted probabilities are used as weights in the regression of

interest. Using a similar approach, in this context, it is investigated if potential selection issues lead to substantial bias in the estimates for married/single women. First, the predicted probability of being married/single is computed on the full sample; then equation (1) is run for married/single mothers using the IPW. Results (not shown) are substantially equal to the estimates shown in Panel B and Panel C of Table 3 (see next Section), confirming that selection bias is not an issue in this context.

Finally, to rule out that our findings are driven by the static DID specification adopted, a specification is employed that controls for the dynamics of the response of divorce rate (and the other marital status rates) and fertility rate to changes in child custody assignment and unilateral divorce. The results (available on request) show that only for separated mothers does the timing of the introduction of joint custody law seems to matter and the effect is increasing over time; however, the other marital statuses and fertility do not show a dynamic response to legal changes.

#### 6.2 Benchmark estimates

The effect of the joint custody reform on the set of mothers' economic outcomes is almost negligible. This is shown in Panel A of Table 3 where the results for mothers are reported after controlling for state fixed- and year-fixed effects and linear state specific time trends.<sup>19</sup> This finding is consistent with the Coase theorem prediction according to which if the costs of bargaining are very low, the switching to joint parenting should have no effect on parental outcomes, since the husband and the wife simply bargain "in the shadow of the law" (Becker 1993; Mnookin and Kornhauser 1979).

Nevertheless, the total effect might hide important heterogeneity. As mentioned by Lundberg and Pollak (1993), for married couples, far from the margin of divorce, changes in legal rules concerning custody arrangements might have little impact on behavior, while divorced or separated couples who have negotiated or are in the process of negotiating the terms of the dissolution of their marriages might be more sensitive to changes in legal rules. Therefore, the effects of the reform might differ according to the family structure.

A separate analysis by marital status might be flawed if the joint-custody reform alters the composition of the married/single population, an effect driven by the potential selection into marriage and divorce (Halla 2013). In this case, it would be difficult to identify the channels through which joint custody operates (Teng 2008). However, evidence provided in Section 6.1 has shown that selection effects into and out of marriage do not represent an issue in this context and therefore a separate analysis of married and single mothers does not confound the causal channels with a selection effect due to a changing composition of population with children.

The results in Panel B of Table 3 show that there is no effect of the legal reform for *married* mothers living in a state with joint custody law as opposed to married mothers living in a non-adoption state for any of the outcomes considered. Although the effect on married women passes most likely through divorce as mentioned before,

<sup>&</sup>lt;sup>19</sup>In the interest of preserving space, only the coefficients on joint custody are reported.

	Poverty	Log earnings	Log other income	Other income (0/1)	On welfare	LFP
	(1)	(2)	(3)	(4)	(5)	(6)
Panel A: All mothers						
joint custody	0.014*	-0.029	0.062	0.007	0.002	0.009
	(0.007)	(0.019)	(0.060)	(0.004)	(0.005)	(0.011)
Ν	1,441,684	842,615	372,553	1,441,684	1,194,473	1,441,684
Panel B: Married mothers						
joint custody	0.011	-0.016	0.003	0.006	0.001	0.013
	(0.007)	(0.019)	(0.078)	(0.004)	(0.002)	(0.011)
Ν	1,213,595	672,480	291,213	1,213,595	979,303	1,213,595
Panel C: Single mothers						
joint custody	0.031**	-0.081***	0.143***	0.009	0.009	-0.007
	(0.012)	(0.024)	(0.039)	(0.008)	(0.019)	(0.014)
Ν	228,089	170,135	81,340	228,089	215,170	228,089

 Table 3
 The effect of joint custody law on mother's labor market outcomes

Source. The 1960, 1970, 1980, 1990, and 2000 Integrated Public Use Microdata Series (IPUMS).

Notes: Control variables include age, race, ethnicity (a dummy variable for Hispanic origin), family structure indicator variables, number of children at home, educational attainment indicator variables (high school graduate, some college and college) for mothers; sex, age, and year of birth of the youngest child in the household. All the regressions include also a full set of state and year dummies, and child age by year interactions. Standard errors are clustered by state of residence. Symbols: \*\*\* significant at 1%; \*\* significant at 5%, \* significant at 10%

both Halla (2013) and the analysis presented in Section 6.1 have proved that the joint custody reform does not affect divorce rates; this might explain the absence of any effects for married mothers. Some of these findings are consistent with results in Altindag et al. (2015), who show that custody reform is unrelated to married mothers' LFP, although it increases the time allocated to market work. Other studies show different results: Nunley and Seals (2011) show that joint-custody reform is associated to higher rates of LFP for married mothers, while Halla (2013) finds that married women are less likely to participate in the labor market in states that adopt joint-custody law. The difference among the two studies might be attributed to the choice of the time period and the dataset employed.

Results for *single* mothers reported in Panel C of Table 3 show that they experience a decrease in their living standards. There is a significant deterioration in earnings (-8.1%) and an increase in the probability of living in poverty (3.1 percentage points), while the amount of other income received increases (14.3%). Moreover,

the introduction of joint custody does affect neither the probability of being in the labor force of single mothers nor their probability of relying on welfare with respect to single mothers living in state without joint custody law. These results are consistent with Brinig and Buckley (1998) who suggest that single mothers might exchange property and child support payments for sole custody, thus making them poorer.

Given that the results show that joint custody has influenced only single mothers, the rest of the analysis will focus on this group of mothers (results for married mothers are available on request). It can be argued that single mothers, who can rely on higher "other income," may have reduced the number of hours worked and consequently their earnings but have not changed their labor market participation. However, the increase in the amount received on "other income" is not enough to compensate the decrease in earnings, increasing then the risk of living in poverty.

As mentioned in Section 5, the amount of exposure to joint custody law may matter and a static approach like the one used in Eq. 1 might not capture the dynamic adjustments of the outcomes to the reform. To explore this issue a variable measuring the exposure to the joint custody law and estimating the effect of joint custody adoption by duration of exposure is constructed. Specifically, the adoption states are divided into the following year intervals after the regime shift: 1–2 years, 3–6 years, 7–10 years, 11–14 years, and 15 and more, as in Eq. 2. In this way, it is possible to take advantage of the fact that individuals interviewed in the same year in different states may have different years of exposure to joint child custody law according to when the individual's state of residence enacted the law. The model is augmented including leads, as indicated in Eq. 3, to capture anticipatory effects in the spirit of Granger (1969). The main goal of the Granger test is to exclude that, conditional on state and year effects, future changes in the law may predict the current outcomes (Autor 2003; Angrist and Pischke 2009). Specifically, three leads are included, for years 1, 2, and 3 before the reform.

Results reported in Table 4 show no evidence of anticipatory effects. The only exception is represented by the variables "being on welfare" and "LFP", for which an anticipatory effect at year t-2 seems at work. However, there is no effect at year t-1, making it harder to believe that the effect at t-2 can be considered as an anticipatory response. When looking at the dynamic adjustments after the adoption of the law, it appears that years of exposure do not matter for poverty, earnings, and probability of receiving other income. For the variables "amount of other income" and "being on welfare," the first 2 years after the joint custody reform are the most important but the effect vanishes afterwards (with more persistence for the amount of other income). Years of exposure turn out to be important only for the variable LFP. When allowing the effect of the reform to vary by years of exposure, the LFP turns out to be significant and approximately constant over time (between about -4.0 and -5.0 percentage points) while in the static case it was negative and not significant. The negative sign is compatible with the hypothesis that single mothers might reduce their time spent parenting and use the additional time on the re-marriage market, thus increasing their probability of re-marriage and in turn decreasing the labor market participation. This explanation has been suggested by Halla (2013), who shows that women in all age groups (except women between 45 and 54 years of age) reduce their labor market participation as a consequence of the reform, although in his case the

	Poverty	Log earnings	Log other income	Other income (0/1)	On welfare	LFP
	(1)	(2)	(3)	(4)	(5)	(6)
Joint custody in effect for:						
year -3	-0.012	0.054	0.015	0.003	-0.010	-0.020
•	(0.019)	(0.046)	(0.102)	(0.010)	(0.012)	(0.012)
year -2	0.013	-0.081	0.084	-0.005	0.030**	-0.030**
	(0.021)	(0.067)	(0.114)	(0.012)	(0.015)	(0.014)
year -1	-0.015	0.097*	0.135	-0.002	-0.027	-0.007
	(0.019)	(0.053)	(0.084)	(0.012)	(0.021)	(0.012)
years 1-2	0.028	-0.044	0.324***	0.012	0.053***	-0.052***
	(0.017)	(0.051)	(0.098)	(0.011)	(0.015)	(0.010)
years 3-6	0.016	0.005	0.202**	0.010	0.018	-0.043***
	(0.016)	(0.050)	(0.090)	(0.013)	(0.015)	(0.012)
years 7-10	0.016	0.018	0.219**	0.016	0.012	-0.044***
	(0.019)	(0.060)	(0.103)	(0.014)	(0.015)	(0.014)
years 11-14	0.019	0.000	0.184	0.018	0.005	-0.049***
	(0.024)	(0.060)	(0.136)	(0.017)	(0.019)	(0.016)
years 15+	0.002	-0.022	0.120	0.014	0.004	-0.040 **
	(0.024)	(0.060)	(0.148)	(0.020)	(0.019)	(0.018)
Ν	228,089	170,135	81,340	228,089	215,170	228,089

 Table 4
 Dynamic DID estimates of the effect of joint custody law on single mothers' labor market outcomes

Source. The 1960, 1970, 1980, 1990, and 2000 Integrated Public Use Microdata Series (IPUMS).

Notes: The sample is made of single mothers. Control variables include age, race, ethnicity (a dummy variable for Hispanic origin), family structure indicator variables, number of children at home, educational attainment indicator variables (high school graduate, some college and college) for mothers; sex, age, and year of birth of the youngest child in the household. All the regressions include also a full set of state and year dummies, and child age by year interactions. Standard errors are clustered by state of residence. Symbols: \*\*\* significant at 1%; \*\* significant at 5%, \* significant at 10%

effect is increasing (in absolute term) over time.<sup>20</sup> Overall, taking into account the dynamic effect of the policy intervention does not seem to provide additional information with respect to the static approach based on the assumption of an immediate response for most of the outcomes under analysis. For simplicity, in what follows the static approach will continue to be used.

 $<sup>^{20}</sup>$ The different pattern of the results between this study and Halla (2013) can be easily reconciled. If one computes the difference between the labor force participation of all women and the labor force participation of married women reported in Table 6 of Halla (2013), this gives the effect for single women which is approximately constant over time.

In addition to investigating the effect of joint custody by duration of exposure, it is also possible to allow the effect of the reform to differ according to the age of the youngest child in the household. The focus is on the youngest child in the household because most of the bargaining between spouses on children custody takes place when children are young and sole custody might be considered as preferred. This analysis might be problematic if joint custody has a direct effect on fertility rates, thus generating a selection effect. However, Section 6.1 has proved that the introduction of joint custody did not have an impact on fertility behavior and therefore the selection problem can be neglected.

The heterogeneity of the effect according to the age of the youngest child in the household at the time of the reform is studied by estimating the following equation:

$$y_{ist} = \mu_s + \tau_t + \mu_s \times t_t + \gamma X_{ist} + \sum_j \beta_j \mathbf{I} \{ JC_{st} = 1 \}$$
$$\times \mathbf{I} \{ a_j \le (YB_{ist} - YJC_s) \le b_j \} + \varepsilon_{ist}$$
(5)

where *YB* is the year of birth of the child *i* whose single mother is living in state *s* at time *t*,  $YJC_s$  is the year of adoption of joint custody law in state *s*; their difference

	Poverty	Log earnings	Log other income	Other income (0/1)	On welfare	LFP
	(1)	(2)	(3)	(4)	(5)	(6)
Joint custody in effect for:						
Born after the reform	0.003	-0.002	0.109	0.005	0.007	0.006
	(0.015)	(0.037)	(0.081)	(0.005)	(0.006)	(0.016)
Age $0-3$ at the time of reform	0.013	-0.022	0.081	0.009*	0.004	0.009
	(0.010)	(0.025)	(0.058)	(0.005)	(0.005)	(0.013)
Age $4-7$ at the time of reform	0.015*	-0.026	0.071	0.009**	0.004	0.008
	(0.008)	(0.021)	(0.069)	(0.004)	(0.005)	(0.012)
Age $8-11$ at the time of reform	0.015**	-0.038*	0.062	0.004	0.002	0.007
	(0.007)	(0.019)	(0.060)	(0.005)	(0.004)	(0.011)
Age 12+ at the time of reform	0.017**	-0.031	0.043	0.007*	-0.000	0.009
	(0.007)	(0.019)	(0.078)	(0.004)	(0.004)	(0.009)
Ν	228,089	170,135	81,340	228,089	215,170	228,089

Table 5 D	DID estimates	by age	of the	child at	joint	custody	reform
-----------	---------------	--------	--------	----------	-------	---------	--------

Source. The 1960, 1970, 1980, 1990, and 2000 Integrated Public Use Microdata Series (IPUMS).

Notes: The sample is made of single mothers. Control variables include age, race, ethnicity (a dummy variable for Hispanic origin), family structure indicator variables, number of children at home, educational attainment indicator variables (high school graduate, some college and college) for mothers; sex, age, and year of birth of the youngest child in the household. All the regressions include also a full set of state and year dummies, and child age by year interactions. Standard errors are clustered by state of residence. Symbols: \*\*\* significant at 1%; \*\* significant at 5%, \* significant at 10%

represents the child's age at the time of the adoption of joint custody law; **I** is an indicator function. The effect of interest is  $\beta_j$  which represents the effect of the move to joint custody for single mothers whose child's age at the time of the reform was in the range  $[a_j, b_j]$  with respect to single mothers living in states that did not adopt joint custody (*JC*=0).  $\beta_j$  is identified using three different sources: the difference in the timing of adoption on joint custody law, the fact that not all the states adopted it, and the difference in child age at time of the reform.

Results are reported in Table 5. All the coefficients for children born after the introduction of the joint custody law are not statistically different from zero. Interestingly, the odds of living in poverty for single mothers increases as the child ages by almost 2 percentage points when the child is 12 years or older. Moreover, a larger fraction of single mothers receive "other income" (0.9 percentage points) when the youngest child is below 7 years of age, although there is no differential effect of the reform by the age of the youngest child on the amount of "other income received."

#### 6.3 The impact of unilateral divorce, property, and child support laws

During the period under analysis, in addition to the unilateral divorce law, many states changed the rules governing property division and removed fault as a consideration in property division (Ellman and Lohr 1998).<sup>21</sup> From a financial point of view, the introduction of new laws regarding property division in a state can be as important as joint custody for a household's wealth after divorce. Moreover, several child support laws to enforce parental payment were passed in the same years. The timing of the introduction of these laws is reported in Appendix Table 8.

In order to identify the net effect of joint custody on single mother's outcomes, changes in family laws concerning unilateral divorce, property division, and child support enforcement are controlled for, as presented in Table 6. The importance of controlling for contemporaneous legal changes has been highlighted by González-Val and Marcén (2012) who have explored the response of the divorce rate to the introduction of the unilateral divorce law once the changes in custody law and Child Support Enforcement program have been controlled for.

For the sake of comparison, column 1 of Table 6 reports the coefficients of Table 3 Panel C; column 2 introduces a dummy variable equal to one if the state *s* has already

<sup>&</sup>lt;sup>21</sup>States can be divided into three groups according to the property division regime: common law property, community property, and equitable division. In a common law regime, marital property is divided at divorce according to who has legal ownership of the property; in a community property regime, most property acquired during the marriage (except for gifts or inheritances) is owned jointly by both spouses and division of community property under divorce may take place by item, by splitting all items or by value; finally, in a equitable division regime, judges have discretion in allocating marital property according to what the judge deems as fair. In equitable division states, judges can act more discretionally in property allocation than in common law or community property states (Gray 1998). Moreover, community property states are considered as transferring more assets to the wife after divorce than in common-law jurisdictions since household assets tend to be held in most cases in the husband's name (Gray 1998).

Table 6 Some robustness checks: the effects of property and child support laws	ustness check.	s: the effects o	f property and cl	hild support law.	S					
	Poverty					Log earnings				
	(1)	(2)	(3)	(4)	(5)	(9)	(1)	(8)	(6)	(10)
Joint custody	$0.031^{**}$	$0.030^{**}$	0.035**	$0.036^{**}$	0.054**	$-0.081^{***}$	$-0.079^{***}$	-0.083***	-0.083 * * *	$-0.110^{**}$
	(0.012)	(0.013)	(0.014)	(0.015)	(0.026)	(0.024)	(0.024)	(0.029)	(0.029)	(0.053)
Unilat. divorce		0.017	0.018	0.018	$0.199^{***}$		-0.053	-0.054	-0.052	$-0.554^{***}$
		(0.024)	(0.023)	(0.023)	(0.061)		(0.050)	(0.050)	(0.049)	(0.124)
No fault			-0.028	-0.028	$-0.073^{***}$			0.020	0.020	$0.168^{***}$
			(0.025)	(0.025)	(0.019)			(0.052)	(0.053)	(0.042)
Equitable div.			$-0.091^{***}$	$-0.086^{***}$	$0.326^{***}$			$0.700^{***}$	$0.672^{***}$	$-0.341^{***}$
			(0.015)	(0.020)	(0.058)			(0.054)	(0.060)	(0.126)
Legislative index				-0.004	-0.001				0.015	0.002
				(0.007)	(0.007)				(0.013)	(0.015)
Child support					0.100					-0.243
					(0.078)					(0.174)
Ν	228,089	228,089	228,089	228,089	191,120	170,135	170,135	170,135	170,135	144,659
	Log other income	ncome				Other income (0/1)	: (0/1)			
	(1)	(2)	(3)	(4)	(5)	(9)	(1)	(8)	(6)	(10)
Joint custody	$0.143^{***}$	$0.143^{***}$	$0.092^{**}$	$0.092^{**}$	0.040	0.009	0.008	0.003	0.003	-0.002
	(0.039)	(0.038)	(0.045)	(0.045)	(0.066)	(0.008)	(0.008)	(0.00)	(0.009)	(0.008)
Unilat. divorce		-0.050	-0.055	-0.056	0.084		0.019	0.018	0.018	0.061
		(0.158)	(0.141)	(0.141)	(0.156)		(0.020)	(0.018)	(0.018)	(0.036)
No fault			$0.217^{***}$	$0.217^{***}$	-0.032			$0.027^{**}$	$0.027^{**}$	-0.013
			(0.080)	(0.079)	(0.076)			(0.012)	(0.012)	(0.011)

Table 6 (continued)	(p									
Equitable div.			$-1.338^{***}$	$-1.335^{***}$	$0.950^{***}$			$-0.101^{***}$	-0.099***	$-0.549^{***}$
			(0.049)	(0.054)	(0.322)			(0.006)	(0.00)	(0.031)
Legislative index				-0.003	0.009				-0.002	0.002
				(0.021)	(0.026)				(0.004)	(0.004)
Child support					-0.358* (0.190)					$0.066^{**}$ (0.030)
Ν	81,340	81,340	81,340	81,340	52,620	228,089	228,089	228,089	228,089	191,120
	On welfare	0				LFP				
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)
Joint custody	0.009	0.010	0.019	0.019	$0.046^{**}$	-0.007	-0.007	-0.011	-0.011	-0.013
	(0.019)	(0.018)	(0.014)	(0.015)	(0.018)	(0.014)	(0.014)	(0.013)	(0.014)	(0.014)
Unilat. divorce		0.020	0.022	0.021	0.015		-0.007	-0.008	-0.007	0.002
		(0.033)	(0.033)	(0.031)	(0.046)		(0.015)	(0.015)	(0.015)	(0.055)
No fault			$-0.050^{**}$	$-0.050^{**}$	$-0.074^{***}$			0.021	0.021	$0.027^{**}$
			(0.024)	(0.024)	(0.016)			(0.014)	(0.013)	(0.010)
Equitable div.			$0.341^{***}$	$0.354^{***}$	$0.237^{***}$			$-0.193^{***}$	$-0.202^{***}$	$-0.424^{***}$
			(0.017)	(0.021)	(0.049)			(0.012)	(0.016)	(0.041)
Legislative index				-0.007	-0.005				0.007	0.009*
				(0.006)	(0.007)				(0.005)	(0.005)
Child support					0.107*					-0.001
					(0.054)					(0.042)
Ν	215,170	215,170	215,170	215,170	191,120	228,089	228,089	228,089	228,089	191,120
Source. The 1960, 1970, 1980, 1990, and 2000 Integrated Public Use Microdata Series (IPUMS)	970, 1980, 1	990, and 2000	Integrated Public	c Use Microdata	Series (IPUMS).					
Notes: The sample is made of single mothers. Control variables include age, race, ethnicity (a dummy variable for Hispanic origin), family structure indicator variables,	is made of si	ingle mothers.	Control variable	s include age, r	ace, ethnicity (a c	lummy variab	ole for Hispan	ic origin), famil	y structure indic:	ttor variables,
number of children at home, educational attainment indicator variables (high school graduate, some college, and college) for mothers; sex, age, and year of birth of the	at home, edu	actional attain	nment indicator v	variables (high s	chool graduate, s	ome college,	and college)	for mothers; sex	, age, and year c	of birth of the
youngest child in the household. All the regressions include also a full set of state and year dummies, and child age by year interactions. Standard errors are clustered by	e household.	All the regres	youngest child in the household. All the regressions include also a full set of state and year	so a full set of st	ate and year dun	mies, and chi	lid age by yea	rr interactions. St	tandard errors are	e clustered by

1226

state of residence. Symbols: \*\*\* significant at 1%; \*\* significant at 5%, \* significant at 10%

🖄 Springer

enacted *unilateral divorce* at t,<sup>22</sup> in column 3 *No-fault Property* and *Equitable Division* dummy variables are added, indicating the presence of specific property division laws.<sup>23</sup> As shown in columns 1–3 of Table 6, adding these controls do not alter the coefficients on joint custody for any of the outcomes considered. Interestingly, living in a equitable property states, where judges have more discretion in allocating marital property, has a direct positive effect on single mothers' earnings, it increases the probability of relying on welfare and reduces the probability of living in poverty, the probability and the amount of income from other sources and the probability of labor market participation.

In the same decades, many states have also been strengthening child support enforcement over time. Since there is a close link between the choice of custody arrangement and child support enforcement, column 4 of Table 6 includes a measure of the child support legislation. Specifically, a single legislative index has been created by summarizing the eight measures of child support legislation mentioned in Section 4, with values ranging from zero for states with no law to eight for states with all eight laws enforced.<sup>24</sup>As before, the results on joint custody are very robust to the inclusion of this measure of child support enforcement.

Similar to González-Val and Marcén (2012), a measure of effectiveness in the collection of child support is built using the Current Population Survey-Child Support Supplement (CPS-CSS) from 1992 to 2000, a nationally representative sample administered biannually by the U.S. Census Bureau. The effectiveness measure is defined as child support collection amount in a state, divided by child support amount owed in the same state. Unfortunately, these data are available only from 1992 and by extrapolating the data back to 1990, the analysis can be run only over the last two decades, 1990 and 2000. For this reason, the results are not fully comparable to the previous ones and must be interpreted with caution. Once the effectiveness in the collection rate is controlled for, the results show that the introduction of the law on joint custody makes matters worse for single mothers since it increases the probability of being below the poverty line and of relying on welfare, it decreases earnings and the probability of being in the labor force (see column 5 of Table 6). Since in most cases, the coefficient on child support effectiveness rate is not significant, the difference in the joint custody coefficient between column 4 and column 5 should be attributed to the different sample size.

<sup>&</sup>lt;sup>22</sup>The divorce coding proposed by Gruber (2004) has been used, but the results are virtually unchanged if the one proposed by Friedberg (1998) is instead used. The main difference between the two is that Gruber considers as "non-adopting" the states that have unilateral divorce and separation requirements (New Jersey, Ohio, Pennsylvania, South Carolina, and the District of Columbia). Results available upon request.

<sup>&</sup>lt;sup>23</sup>Three states (Massachusetts, Missouri and Montana) shifted from common law property division to equitable division over the period. Equitable division is included as control, common law is the excluded category and community property is omitted because of collinearity with state fixed effects.

<sup>&</sup>lt;sup>24</sup>The results are unchanged if all the eight measures are included instead of the single index or if the single index is computed through a factor analysis and taking the first principal component. Results available upon request.

## 6.4 A threat to the identification strategy

A drawback of the identification strategy is that substantial interstate migration in response to changes in joint custody laws might bias the estimates, as mentioned in Section 5. In other words, individuals might endogenously move in view of states' changes in family law, introducing estimation bias. It can be argued that endogenous migration is not a problem in this context for several reasons.

First, migration of parents during their children's teenage years is not very widespread. In the sample of single mothers, at the survey year only 14.3% of children live in a state different from the state of birth. Second, if parents fully agree on shared custody at the time of divorce they do not need to migrate to another state with a more favorable law on joint custody to enforce the agreement. On the contrary, if parents do not agree on the form of custody arrangement, the primary custodial parent cannot move with the child to another state without the permission of the other parent because child abduction is considered a criminal offence. Finally, when the custody arrangement is settled, the custodial parent, usually the mother, who wishes to relocate might be prohibited by the court from doing so unless she is willing to give up the custody of her child(ren) (Folberg 1991, p.84).

A simple way to assess the severity of bias induced by potential selective migration is to use the change in law in the child's state of birth (SOB) instead of the state of residence (SOR) and evaluate the sensitivity of the results. This check captures the case of a woman who changes residence after divorce to match a more favorable custody law.<sup>25</sup> The estimates using either "change of joint custody law in the SOR" or "change of joint custody law in the SOB" turn out to be similar. As additional robustness check, Eq. 1 is estimated for single mothers excluding those whose child's place of birth is different from the current state of residence to evaluate the importance of potential endogenous movers. The results are again similar to what shown in Table 3 Panel C, confirming that endogenous migration is not affecting the estimates (both set of results are available upon request).

## 7 Interpretation of the results

The analysis above has shown that the adoption of the joint custody law affects single mothers in terms of reduced earnings, increased probability of living in poverty and increased amount of "other income" received. In this section, a potential explanation for these findings is provided by exploring the link between joint child custody and child support. For this purpose, the analysis uses the Current Population Survey-Child Support Supplement (CPS-CSS) from the years 1994 to 2002. Information

<sup>&</sup>lt;sup>25</sup>It is assumed that children were born before the divorce took place, which seems a reasonable assumption.

gathered from respondents includes data on custody arrangements, child support payments, and visitation, and detailed individual and family characteristics. The sample of women selected are between 14 and 60 years of age, with their own children below 18 years in the household, who are supposed to receive child support payments. The dependent variable is the collection rate computed as the ratio between the amount of child support actually received last year over the amount of money supposed to be received as child support.<sup>26</sup> Importantly, in the CPS-CSS, the respondent is asked if a court or judge ever gives the joint legal or physical custody of the child(ren). It is here investigated whether being awarded joint (legal or physical) custody has an effect on the child support payment and in particular on the collection rate.

Table 7 first considers the effect of joint legal custody on collection rate (Panel A), then the effect of joint physical custody (Panel B) and finally both types of custody arrangements simultaneously (Panel C). It also controls for race, education, number of children below 18 years old, marital status, personal income, and being below the poverty line (the last two variables are potentially endogenous but the results are robust to their exclusion). The results show that mothers awarded joint legal custody increase the collection rate by between 7 and 10 percentage points. This result is robust to the model specification (columns 1-5) and to the choice of the sample (for example considering only divorced and separated mothers as in column 6). Moreover, mothers with high school education or more increase significantly the collection rate if awarded joint legal custody (columns 7 and 8), while the effect is not significant for mothers with less that high school (column 9). When we look at Panel B, the effect of being awarded joint physical custody has a positive and significant effect on the collection rate (between 6.0 and 7.0 percentage points). As before, there is some heterogeneity according to the level of education. Interestingly if joint legal and physical custody are controlled for simultaneously (Panel C), only the shared legal custody of the child(ren) matters. This result might be due to the high collinearity of these two variables.<sup>27</sup>

<sup>&</sup>lt;sup>26</sup>All the income variables are expressed in 1999 dollars.

<sup>&</sup>lt;sup>27</sup>Some unobserved characteristics may affect both joint custody and the financial involvement of nonresident parents. Parents who choose joint custody are different from those who do not in terms of disagreement on post-divorce child-rearing decisions or conflictual relationship between partners. Here a 2SLS strategy is used to adjust for such unobserved characteristics in order to have unbiased estimates of joint custody (Seltzer 1991, 1998). Following Seltzer (1998), the average percentages of joint custody in each state are used as an instrument for actual joint custody. The effect of the percentage of state joint custody (either legal or physical) on observed joint custody is of the expected sign and strongly significant. This represents our first stage regression. As one might expect, parents who live in states with a higher percentage of joint custody are more likely to share child custody. Results from the 2SLS regressions are very similar in size to those reported in Table 7 even if mainly not significantly different from zero due to the high standard errors (results available on request). The Hausman test of equality between OLS and IV does not reject the null hypothesis of no differences between OLS and 2SLS estimates.

Table 7 Effect of joint legal and physical custody on the collection rate	al and physical	l custody on the	collection rate						
	All	All	All	All	All	only divor/separ	high school	school	high school
	(1)	(2)	(3)	(4)	(5)	(9)	(1)	(8)	(6)
Panel A									
Joint legal	$0.105^{***}$	0.097 ***	$0.084^{***}$	$0.077^{***}$	$0.074^{***}$	$0.079^{***}$	$0.070^{***}$	0.092 ***	0.048
	(0.007)	(0.008)	(0.008)	(0.008)	(0.008)	(0000)	(0.011)	(0.019)	(0.052)
Panel B									
Joint physical	$0.078^{***}$	$0.072^{***}$	$0.064^{***}$	$0.058^{***}$	0.058 * * *	$0.057^{***}$	0.056***	0.058**	0.019
	(0.00)	(600.0)	(0.008)	(0.00)	(0.008)	(0.011)	(0.011)	(0.022)	(0.068)
Panel C									
Joint legal	0.099***	$0.091^{***}$	$0.078^{***}$	$0.071^{***}$	$0.068^{***}$	$0.074^{***}$	$0.063^{***}$	$0.092^{***}$	0.049
	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)	(0.011)	(0.014)	(0.024)	(0.057)
Joint physical	0.014	0.015	0.016	0.014	0.016	0.012	0.018	-0.000	-0.004
	(0.012)	(0.012)	(0.012)	(0.012)	(0.011)	(0.014)	(0.014)	(0.028)	(0.075)
Race	No	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Children	No	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Education	No	No	Yes	Yes	Yes	Yes	Ι	Ι	Ι
Marital status	No	No	No	Yes	Yes	I	Yes	Yes	Yes
Income-related variables	No	No	No	No	Yes	No	No	No	No
Year fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
State-fixed effects	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Ν	8102	8102	8102	8102	8102	4380	2454	1502	424
Notes: Data from the Current Population Survey-Child Support Supplement (CPS-CSS) from 1994 to 2002. The sample is made by women who were supposed to receive child support payments in the past year. The dependent variable is the collection rate ratio between the amount of child support actually received last year over the amount of money supposed to be received in child support last year. Control variables include race (black, other and white as omitted category), children (one, two, or more than 2 as omitted category), education (less than high school diploma as omitted category), income-related variables (being below the poverty line and more than high school diploma), marital status (separated, divorced, never married, and married as omitted category), income-related variables (being below the poverty line and individual income), plus a full set of year and state dummies. From column 6 to 9, only divorced/separated mothers. Standard errors are clustered by state of residence. Symbols: *** significant at 10%	ent Population the past year. T eceived in chil ication (less th nd married as c o 9, only divor	Survey-Child S The dependent v d support last y an high school pmitted category ced/separated n	upport Suppler ariable is the c ear. Control va diploma as om y), income-relat nothers. Standa	nent (CPS-CSS ollection rate <i>rz</i> riables include itted category, l ted variables (b rd errors are clu	) from 1994 to ttio between the race (black, oth high school dip eing below the istered by state	ulation Survey-Child Support Supplement (CPS-CSS) from 1994 to 2002. The sample is made by women who were supposed to receive i year. The dependent variable is the collection rate ratio between the amount of child support actually received last year over the amount I in child support last year. Control variables include race (black, other and white as omitted category), children (one, two, or more than (less than high school diploma as omitted category, high school diploma and more than high school diploma), marital status (separated, ied as omitted category), income-related variables (being below the poverty line and individual income), plus a full set of year and state ly divorced/separated mothers. Standard errors are clustered by state of residence. Symbols: *** significant at 1%; ** significant at 5%, *	made by women port actually rece tted category, ch high school diplo ividual income), p ls: *** significant	who were suppor ived last year o' lidren (one, two ma), marital sta alus a full set of at 1%; ** signi	sed to receive ver the amount , or more than tus (separated, year and state ficant at 5%, *

The results are consistent with Pearson and Thoennes (1988) who find that joint legal custody increases compliance with child support orders and with Allen et al. (2011) who show that joint-custody enactment raises the probability of receiving child support for single mothers, with an effect mainly driven by divorced mothers, while separated and never-married mothers are unaffected by the joint custody reform. Seltzer and Maralani (2001) also look at the effect of the introduction of joint custody on child support payments received by single mothers and come to different conclusions: there is no evidence of an effect of joint custody law on the probability of receiving child support. The discrepancy between the results in this paper and those of Seltzer and Maralani (2001) might be due to the different time period of analysis, longer in this case, which could capture the increased ability of states to collect child support over time.

The results confirm that in the case of joint (legal or physical) custody mothers can rely on stronger child support payment. This is in line with the arguments made by Weiss and Willis (1985) who suggest that the opportunity of exchanging custody for payment support or other transfers could increase the contribution from the non-custodial parent. However, this belief might have consequences for mothers who, relying on the larger financial involvement of fathers, might be less interested in looking for high-paid jobs or advancing personal careers or might simply reduce the number of hours worked. This behavior might lead to a decrease in mothers' living standards in case the payments from fathers do not compensate the mothers' drop in earnings after divorce. Moreover, as shown in Table 7, joint custody increases the collection rate mainly for highly educated mothers; according to the theory of assortative mating, those mothers are more likely to have a highly educated partner, potentially with higher incomes. If only single mothers with high income nonresident fathers benefit from joint custody, as shown by Seltzer and Maralani (2001), this might explain the negative effect of joint custody on mothers' economic outcomes.<sup>28</sup>

## 8 Summary and Conclusions

The main goal of the adoption of the joint custody law was to make easier the access of the non-custodial parent (usually the father) to their child(ren), to increase the

<sup>&</sup>lt;sup>28</sup>Potentially, the results might also be explained by the fact that joint legal custody places more restrictions on the residential parent than sole custody does. Courts may restrict the geographic area in which a parent lives as part of the custody order, or they may deny a subsequent request for permission to move if it is viewed as an attempt to hinder the other parent's visitation. To shed some light on this channel, this study investigates if divorced/separated mothers are less able to move under joint custody than under sole custody. Using the CPS-CSS data and the same sample of Table 7, it is found that mothers awarded joint custody do have the same probability of moving than mothers with sole custody. Moreover, there is no evidence that joint custody places restrictions on career investment since divorced/separated mothers with joint custody have a higher probability of being employed, earn more and rely less on welfare.

financial support but also to share child-rearing responsibilities between parents after divorce. This paper investigates the effect of the introduction of the joint custody law on mothers' labor market outcomes. It uses the variation occurring from the different timing of custody law reforms across the US states as a source of identification. The results show that the consequences of the adoption of the joint custody law are very different between single and married mothers. Married mothers are not affected by the reform, while single mothers experience a decrease in earnings and an increase in other income received, which however do not compensate each other. As a consequence, single mothers are more likely to live in poverty. The results are robust to controlling for unilateral divorce, specific property division laws and child support legislation. There is no evidence that the exposure to joint custody law affects mothers' outcomes. When the heterogeneity of the effect depending on the age of the youngest child in the household is investigated, the odds of living in poverty are found to increase as the child ages, when child's economic needs tend to increase. The possibility that the adoption of joint custody law has any compositional effects has also been excluded: on aggregate, there is no significant change in the proportion of children living with a single parent or in an intact family. This helps to rule out that the results are driven by a change in marital choices due to the adoption of joint custody law. Using the Current Population Survey-Child Support Supplement 1994-2002 there is a possible explanation for these results: the higher child support payments a mother receives from the non-custodial father in case of joint custody might discourage her from looking for high paid jobs or investing in her careers. However, the increase in compliance with child support payment is not enough to compensate for the reduction in mothers' earnings if it mainly comes from higher income fathers. Finally, the result of an increase in the amount of other income received by single mothers combined with an increase in the collection rate following the joint custody reform potentially suggest the existence of fewer conflicts among separated partners. This result is compatible with findings by Halla (2013) and Wolfers (2006) who show that suicide rates and domestic violence have decreased after the introduction of joint custody. However, the findings also show a decrease in earnings and an increase in the probability of living in poverty. To ascertain if these changes had an overall positive or a negative effect on single mothers' well-being is however left for future research.

**Acknowledgments** I thank Josh Angrist, Erich Battistin, and Sascha Becker for helpful discussions. I also thank two anonymous referees for their help and guidance. The author is responsible for all errors and omissions.

#### **Compliance with Ethical Standards**

Conflict of Interests The author declares that she has no conflict of interest.

# Appendix

					Child sup	port enforce.	Child support enforcement variables	6				
State	Joint custody	Unilateral divorce	No-fault prop. division	Property settl. law	Genetic tests	Paternity test	Numeric. guidelines	Presump. guidelines	Wage withhol.	Imm. withhol.	Univ. withhol.	State interc.
Alabama	1997	1971	Fault	Common law	1984	1984	1987	1989	1984	1989	1993	1985
Alaska	1982	1935	1974	Equi. dist.	1989	1976	1987	1987	1981	1988	1994	
Arizona	1989	1973	1973	Comm. Prop.	1984	1971	1989	1989	1977	1987	1987	1985
Arkansas	2003	0	1979	Equi. dist.	1989	1985	1989	1989	1983	1989	1991	1983
California	1980	1970	1970	Comm. Prop.	1986	1986	1990	1990	1980	1989	1989	1984
Colorado	1987(1983 <sup>a</sup> )	1972	1971	Equi. dist.	1991	1985	1986	1986	1981	1989	1993	
Connecticut	1980	1973	Fault	Equi. dist.	1989	1985	1989	1989	1955	1983	1983	1991
Delaware	1981	1968	1974	Equi. dist.	1984	1984	1983	1988	1974	1990	1993	1992
D.C.	1996	0	Fault	Equi. dist.	1984	1984	1990	1990	1987	1990	1995	1987
Florida	1982	1971	1986	Common law	1986	1986	1987	1989	1978	1986	1986	
Georgia	1990	1973	Fault	Common law	1991	1980	1989	1989	1981	1989	1993	1980
Hawaii	1980	1972	1960	Equi. dist.	1989	1983	1986	1986	1984	1988	1988	1982
Idaho	1982	1971	1990	Comm. Prop.	1982	1985	1989	1989	1986	1990	1993	1981
Illinois	1986	0	1977	Equi. dist.	1984	1984	1984	1990	1984	1988	1988	1989
Indiana	1983	1973	1973	Equi. dist.	1987	1986	1989	1989	1982	1985	1997	1981
Iowa	1977	1970	1972	Equi. dist.	1989	1990	1984	1989	1984	1990	1993	1980
Kansas	1980	1969	1990	Equi. dist.	1994	1985	1986	1992	1985	1990	1992	1981
Kentucky	1980	1972	Fault	Equi. dist.	1984	1986	1990	1990	1984	1988	1988	1986
Louisiana	1981	0	Fault	Comm. Prop.	1985	1980	1989	1989	1982	1989	1993	1992

Table 8 Year of introduction of joint custody, divorce laws, and child support enforcement variables and types of marital property laws

					Child sup	port enforce	Child support enforcement variables	SS				
State	Joint custody	Unilateral divorce	No-fault prop. division	Property settl. law	Genetic tests	Paternity test	Numeric. guidelines	Presump. guidelines	Wage withhol.	Imm. withhol.	Univ. withhol.	State interc.
Maine	1981	1973	1985	Equi. dist.	1991	1985	1989	1989	1985	1990	1991	1985
Maryland	1986	0	Fault	Common law	1984	1984	1989	1989	1976	1991	1993	1957
Massachusetts	1982	1975	Fault	Common law <sup>c</sup>	1986	1986	1989	1989	1986	1986	1998	1986
Michigan	1981	1972	Fault	Equi. dist.	1982	1986	1985	1990	1982	1990	1990	1985
Minnesota	1982	1974	1974	Equi. dist.	1980	1980	1983	1983	1978	1990	1993	1980
Mississippi	1983	0	Fault	Common law	1987	1981	1989	1989	1985	1989	1993	1985
Missouri	1983	0	Fault	Common law <sup>c</sup>	1987	1987	1989	1989	1973	1990	1993	1984
Montana	1981	1973	1975	Common law <sup>d</sup>	1989	1985	1989	1989	1985	1989	1991	1985
Nebraska	$0(1983^{a})$	1972	1972	Equi. dist.	1984	1986	1985	1985	1985	1991	1994	1984
Nevada	1981	1967	1973	Comm. Prop.	1989	1983	1987	1989	1985	1989	1993	
New Hampshire	1981	1971	Fault	Equi. dist.	1988	1985	1988	1988	1985	1993	1993	
New Jersey	1991	0	1980	Equi. dist.	1983	1983	1986	1988	1981	1990	1990	1985
New Mexico	1982	1933	1976	Comm. Prop.	1986	1986	1988	1989	1985	1990	1993	1985
New York	0	0	Fault	Common law	1976	1985	1989	1989	1977	1990	1994	1985
North Carolina	1988(1957 <sup>a</sup> )	0	Fault	Common law	1979	1981	1985	1989	1975	1989	1993	1979
North Dakota	0	1971	Fault	Equi. dist.	1989	1975	1983	1989	1979	1989	1989	1983
Ohio	1981	0	Fault	Common law	1986	1982	1990	1990	1986	1993	1993	1985
Oklahoma	1983	1953	1975	Equi. dist.	1985	1985	1988	1989	1978	1989	1994	1985
Oregon	$1987(1977^{a})$	1971	1971	Equi. dist.	1981	1983	1989	1989	1985	1993	1993	1985
Pennsylvania	1981	0	Fault	Common law	1989	1985	1985	1989	1985	1989	1989	1985

 Table 8 (continued)

					Child sup	port enforce	Child support enforcement variables	Ş				
State	Joint custody	Unilateral divorce	No-fault prop. division	Property settl. law	Genetic tests	Paternity test	Numeric. guidelines	Presump. guidelines	Wage withhol.	Imm. withhol.	Univ. withhol.	State interc.
Rhode Island	0	1975	Fault	Common law	1984	1988	1987	1987	1980	1990	1994	1982
South Carolina	1996	0	Fault	Common law	1984	1984	1989	1989	1985	1989	1994	1984
South Dakota	1989	1985	Fault	Equi. dist.	1989	1986	1989	1989	1986	1990	1990	
Tennessee	1986	0	Fault	Common law	1983	1984	1989	1989	1985	1990	1994	
Texas	1987	1970	Fault	Comm. Prop.	1989	1983	1989	1989	1985	1985		
Utah	1988	1987	1983	Equi. dist.	1992	1990	1989	1989	1977	1990	1993	1985
Vermont	$0(1987^{a})$	0	Fault	Equi. dist.	1983	1983	1985	1985	1983	1989	1989	1989
Virginia	1987	0	Fault	Common law	1985	1988	1988	1988	1982	1988	1995	1950
Washington	1987 <sup>b</sup>	1973	1973	Comm. Prop.	1994	1976	1988	1988	1984	1989	1994	
West Virginia	0	0	Fault	Common law	1989	1986	1986	1989	1986	1990	1993	1995
Wisconsin	1988(1977 <sup>a</sup> )	1978	1977	Equi. dist.	1987	1983	1983	1987	1977	1985	1989	1987
Wyoming	0	1977	Fault	Equi. dist.	1989	1978	1989	1989	1986	1989	1994	
Notes: 0 in colu propert, Comm. Year of no-fault	mns 1 and 2 dend Law stands for divorce: Ellman	otes that a stat Common Lav and Lohr (199	Notes: 0 in columns 1 and 2 denotes that a state legislation was never adopted or acknowledged. Eq. dist stands for equitable distribution, Comm. Prop. stands for community propert, Comm. Law stands for Common Law. Sources: Unilateral Divorce coding: Gruber (2004); Joint Custody: Wah Leo (2008); Property division types: Gray (1998); Year of no-fault divorce: Ellman and Lohr (1998). Genetic tests, paternity, numerical guidelines, presumptive guidelines, wage withholding, immediate withholding, universal	never adopted or teral Divorce coc paternity, numer	acknowled ling: Grube ical guideli	lged. Eq. dist er (2004); Jo ines, presum	t stands for eq aint Custody: ptive guidelin	uitable distrib Wah Leo (200 es, wage with	oution, Com 08); Propert	m. Prop. sta y division t mediate wit	ands for con types: Gray thholding, u	umunity (1998); niversal
withholding eta	withholding state intercent are from Hugge at al. $(2003)$	rom Huana at	· a1 (2003)	1	1	I						

withholding, state intercept are from Huang et al. (2003)

<sup>a</sup>Consent of both parents required in joint custody cases

 $^{\mathrm{b}}\mathrm{A}\mathrm{greement}$  required, but will not have bearing if found to be unreasonable

<sup>c</sup>Changed to equitable division in 1976

<sup>d</sup>Changed to equitable division in 1974

Table 8 (continued)

# References

- Allen B, Nunley J, Seals A (2011) The Effect of Joint-Child-Custody Legislation on the Child-Support Receipt of Single Mothers. J Fam Econ Iss 32(1):124–139
- Allen DW, Brinig M (2005) Bargaining in the shadow of child custody. University of Iowa Legal Studies Research Paper No
- Allen DW, Brinig M (2011) Do joint parenting laws make any difference? J Empir Leg Stud 8(2):304-324
- Altindag D, Nunley J, Seals A (2015) Child-custody reform and the Division of Labor in the Household. Review of the Economics of the Household. https://doi.org/10.1007/s11150-015-9282-0
- Angrist J, Pischke JF (2009) Mostly harmless econometrics: an empiricist's companion. Princeton University Press, Princeton
- Autor D (2003) Outsourcing at will: unjust dismissal doctrine and the growth of temporary help employment. J Labor Econ 21(1):1–42
- Bauserman R (2002) Child adjustment in joint-custody v sole-custody arrangements: a meta-analytic review . J Fam Psychol 16(1):91–102
- Bartlett KT, Stack CB (1991) Joint custody, feminism, and the dependency dilemma. In: Folberg J (ed) Joint Custody and Shared Parenting. The Guilford Press, New York
- Becker GS (1993) A treatise on the family. Harvard University Press, Cambridge
- Bertrand M, Duflo E, Mullainathan S (2004) How Much Should We Trust Differences-in-Differences Estimates? Q J Econ 119(1):249–275
- Böheim R, Francesconi M, Halla M (2012) Does Custody Law Affect Family Behavior In and Out of Marriage?, IZA dp7064
- Böheim R, Francesconi M, Halla M (2016) Child Custody Law and Family Behavior: Evidence from Austria, mimeo
- Brinig M, Buckley FH (1998) Joint Custody: Bonding and Monitoring Theories. Indiana Law J 73(2):1-31
- Browning M, Chiappori PA (1998) Efficient Intra-Household Allocations: A General Characterization and Empirical Tests. Econometrica 66(6):1241–1278
- Brown P, Melli M, Cancian M (1997) Physical custody in wisconsin divorce cases, 1980-1992, Discussion paper 1113, Institute for research on poverty. University of Wisconsin-Madison, Madison
- Cancian M, Meyer DR (1998) Who gets custody Demography 35(2):147–157
- Chen Y (2013) Is the best interests of the child best for every childthe long-term implications of genderneutral custody laws, Working Paper. Ohio State University, Columbus
- Chiappori PA, Fortin B, Lacroix G (2002) Marriage market, divorce legislation, and household labor supply. J Polit Econ 110(2):37–72
- De Blasio G, Vuri D (2013) Joint Custody in the Italian Courts, IZA dp 7472
- Del Boca D, Ribero R (1998) Transfers in non-intact households. Struct Chang Econ Dyn 9:469-478
- Ellman IM, Lohr SL (1998) Dissolving the Relationship between Divorce Laws and Divorce Rates. Int Rev Law Econ 18(3):341–359
- Folberg J (1991) Joint custody and shared parenting. Guilford Press, New York
- Friedberg L (1998) Did Unilateral Divorce Raise Divorce Rates? Evidence from Panel Data. Am Econ Rev 88(3):608–627
- Genadek KR, Stock WA, Stoddard C (2007) Divorce Laws and the Labor Force Participation of Women with and without Children. J Hum Resour 42(1):247–274
- González-Val R, Marcén M (2012) Unilateral divorce versus child custody and child support in the U.S. J Econ Behav Organ 81(2):613–643
- Gray JS (1998) Divorce-Law Changes, Household bargaining, and Married Women's Labor Supply. Am Econ Rev 88(3):628–642
- Granger CWJ (1969) Investigating causal relations by econometric models and cross-spectral methods. Econometrica 37(3):424–438
- Grossbard-Shechtman S (1995) Marriage market models. In: The New Economics of Human Behavior, by Mariano Tommasi and Kathryn Ierulli, 92-112. New York: Cambridge University Press
- Gruber J (2004) Is Making Divorce Easier Bad for Children? J Labor Econ 22(4):799–833
- Halla M (2013) The effect of joint custody on family outcomes. J Eur Econ Assoc 11(2):278-315
- Huang CC, Han W, Garfinkel I (2003) Child support enforcement, joint legal custody, and parental involvement. Soc Serv Rev 77(2):255–278
- Huang CC, Kunz J, Garfinkel I (2002) The effect of child support on welfare exits and re-entries. J Policy Anal Manage 21(4):557–576

- Huang CC (2002) The impact of child support enforcement on nonmarital and marital births: Does it differ by racial and age groups Soc Serv Rev 76(2):275–301
- Jacob H (1988) Silent Revolution: The Transformation of Divorce Law in the United States. University of Chicago Press, Chicago
- Kuehl SJ (1989) Against joint custody: a dissent to the general bull moose theory. Family and Conciliation Courts Review 27(2):37–45
- Lundberg S, Pollak RA (1993) Separate Spheres Bargaining and the Marriage Market. J Polit Econ 101(6):988–1010
- Maccoby EE, Depner CE, Mnookin RH (1991) Co-parenting in the second year after divorce. In: Folberg J (ed) Joint Custody and Shared Parenting. The Guilford Press, New York
- Maccoby EE, Mnookin RH (1997) Dividing the child: social and legal dilemmas of custody. Harvard University Press, Harvard
- Maiti A (2015) Effect of joint custody laws on childrens future labor market outcomes. Int Rev Law Econ 43:22–31
- Manser M, Brown M (1980) Marriage and Household Decision-Making: A Bargaining Analysis. Int Econ Rev 22(1):31–44
- Mason MA (1999) The custody wars: why children are losing the legal battles and what we can do about it, Basic
- McElroy M, Horney MJ (1981) Nash-bargained household decisions: toward a generalization of the theory of demand. Int Econ Rev 22(2):333–349
- Mnookin RH, Kornhauser L (1979) Bargaining in the Shadow of the Law: The Case of Divorce. Yale Law J 88(5):950–997
- Morrow AM (1991) Shared physical custody: economic consideration. In: Folberg J (ed) Joint Custody and Shared Parenting. The Guilford Press, New York
- Nunley JM, Seals RA (2011) Child-custody reform, marital investment in children, and the labor supply of married mothers. Labour Econ 18(1):14–24
- Pearson J, Thoennes N (1988) Divorce mediation research results. In: Folberg J, Milne A (eds) Divorce Mediation Theory and Practice, New York, Guilford Press
- Rasul I (2006) The economics of child custody. Economica 73(1):1-25
- Rose E, Wong C (2014) But who will get billy? the effect of child custody laws on marriage, IZA DP No. 8611
- Ruggles S, Sobek M (1997) Integrated public use microdata series: Version 2.0. Minneapolis: University of Minnesota, Historical Census Projects
- Seltzer J (1991) Legal Custody Arrangements and Children's Economic Welfare. Am J Sociol 96(4):895– 929
- Seltzer J (1998) Father by Law: Effects of Joint Legal Custody on Nonresident Fathers' Involvement with Children. Demography 35(2):135–46
- Seltzer JA, Maralani VJ (2001) Joint legal custody and child support payments: are there lasting custody effects?, California Centre for Population Research (CCPR) 004–01
- Stevenson B (2007) The impact of divorce laws on marriage-specific capital. J Labor Econ 25(1):75–94
- Stevenson B (2008) Divorce law and women's labor supply. J Empir Leg Stud 5(4):853-873
- Stevenson B, Wolfers J (2006) Bargaining in the shadow of the law: divorce laws and family distress. Q J Econ 121(1):267–288
- Teng WL (2008) From maternal preference to joint custody: the impact of changes in custody law on child educational attainment, mimeo
- Weiss Y, Willis RJ (1985) Children as collective goods and divorce settlements. J Labor Econ 3(3):268– 292
- Wolfers J (2006) Did Unilateral Divorce Raise Divorce Rates? A Reconciliation and New Results. Am Econ Rev 96(5):1802–1820
- Wooldridge J (2002) Inverse probability weighted M-estimators for sample selection, attrition, and stratification. Port Econ J 1(2):117–139
- Wooldridge J (2003) Econometric Analysis of Cross Section and Panel Data. The MIT Press, Cambridge